Author’s response to reviews

Title: Pathogenetic approach to the treatment of functional disorders of the gastrointestinal tract and their intersection: Results of the Russian Observation Retrospective Program COMFORT

Authors:

Vladimir Ivashkin (kont07@yandex.ru)
Elena Poluektova (polouektova@rambler.ru)
Alexey Glazunov (makarovaea@materiamedica.ru)
Mikhail Putilovskiy (PutilovskiyMA@materiamedica.ru)
Oleg Epstein (EpsteinOI@materiamedica.ru)

Version: 2 Date: 22 Nov 2019

Author’s response to reviews:

Dear Dr. Jashan Abraham,

First of all let me thank you for your valuable comments.

Please find the attached documents that were reworker in accordance with your comments. We information concerning the ethical approval was detalised and the role of founders was mentioned in the Funding section. The title page supplementary material was removed and added before the Abstract section. The manuscript is clean without highlights or comments. Thanks again, Hope the article will have a chance to be published in the BMC gastro. Looking forward to hearing from you, Kind regards, Prof. Glazunov